Author's response to reviews

Title: Smad4-expression is decreased in breast cancer tissues: a retrospective study

Authors:

Christina H Stuelten (chrisstu@mail.nih.gov)
Miriam B Buck (miriam.buck@ikp-stuttgart.de)
Juergen Dippon (juergen.dippon@mathematik.uni-stuttgart.de)
Anita B Roberts (Robertsa@mail.nih.gov)
Peter Fritz (peter.fritz@rbk.de)
Cornelius Knabbe (cornelius.knabbe@rbk.de)

Version: 3 Date: 12 January 2006

Author's response to reviews: see over
To whom it may concern:

As requested we made the following changes in the manuscript:

1) We removed the header from the manuscript
2) We removed the line numbers
3) We removed the titles from the images
4) We removed the additional files
5) We controlled Title, Abstract, and Author list.

Sincerely

Christina Stuelten

Laboratory of Cell Regulation and Carcinogenesis
NCI, NIH
41 Library Drive, MSC 5055
Bethesda, MD 20892-5055
Phone: 301-496-0693
Fax: 301-496-8395
e-mail: chrisstu@mail.nih.gov